Author’s response to reviews

Title: Guidelines for randomized clinical trial protocol content: a systematic review

Authors:

Jennifer M Tetzlaff (jtetzlaff@ohri.ca)
An-Wen Chan (anwen.chan@utoronto.ca)
Jessica Kitchen (jessica.kitchen@wchospital.ca)
Margaret Sampson (msampson@cheo.ca)
Andrea Tricco (triccoa@smh.ca)
David Moher (dmoher@ohri.ca)

Version: 2 Date: 26 March 2012

Author’s response to reviews:

Editors-in-Chief Systematic Reviews Journal
March 26, 2012

Dear Editors-in-Chief and colleagues,

This is a resubmission of the manuscript entitled: “Guidelines for randomized clinical trial protocol content: a systematic review”.

Please find the completed PRISMA checklist attached with this submission.

We look forward to your response.

Best wishes,

Jennifer M. Tetzlaff
Ottawa Methods Centre, Clinical Epidemiology Program, Ottawa Hospital Research Institute
501 Smyth Rd., Ottawa, Ontario, Canada, K1H 8L6
Telephone: 613-595-1959, Email: jen_tetzlaff@yahoo.ca